Author's response to reviews

Title: In vitro antifungal activity and cytotoxicity effects of fresh xylem sap of Hymenaea courbaril L. and its major constituent fisetin

Authors:

Maysa P Costa Costa, MP (maysa_paula@yahoo.com.br)
Marize CV Bozinis Bozinis, MCV (marizecv@farmacia.ufg.br)
Wanessa M Andrade Andrade, WM (wanessa.essa@gmail.com)
Carolina R Costa Costa, CR (carolrc80@yahoo.com.br)
Cecilia M Alves Cecilia, MA (cecilia@quimica.ufg.br)
Lucilia Kato Kato, L (lucilia@quimica.ufg.br)
Orionalda FL Fernandes Fernandes, OFL (orionlisboa51@gmail.com)
Lucia KH Souza Souza, LKH (luciakioko@ufg.br)
Maria do Rosario R Silva Silva, MRR (rosario@ufg.br)

Version: 2
Date: 27 November 2013

Author's response to reviews: see over
To: Editor-in-Chief

BMC Complementary and Alternative Medicine

From

Profa. Dra. Maria do Rosário Rodrigues Silva

We are sending the manuscript “In vitro antifungal activity and cytotoxicity effects of fresh xylem sap of *Hymenaea courbaril* L. and its major constituent fisetin” with the suggested corrections by editor for publication in BMC Complementary and Alternative Medicine. We reorganized Methods and added “Extraction and Isolation” and “Structure elucidation” as suggested. Details of quantity of fisetin and compound mixture are showed in page 3: *fisetin* (1, 10mg) and 30 mg of the mixture of the compounds *fisetinediol* (2); *fustin* (3); 3-O methyl, 2, 3-\(E\)-*fustin* (4) and *taxifolin* (5)

We hope to contribute to increasing the knowledge of medicinal plants in use mainly as antifungal.

Best regards

Profa. Dra. Maria do Rosário R. Silva